import { TypeList } from "docs-ui"

# fulfillment

## Interfaces

- [AdjustmentLineDTO](../../fulfillment/interfaces/fulfillment.AdjustmentLineDTO/page.mdx)
- [BaseFilterable](../../fulfillment/interfaces/fulfillment.BaseFilterable/page.mdx)
- [CalculateShippingOptionPriceDTO](../../fulfillment/interfaces/fulfillment.CalculateShippingOptionPriceDTO/page.mdx)
- [CartAddressDTO](../../fulfillment/interfaces/fulfillment.CartAddressDTO/page.mdx)
- [CartCreditLineDTO](../../fulfillment/interfaces/fulfillment.CartCreditLineDTO/page.mdx)
- [CartDTO](../../fulfillment/interfaces/fulfillment.CartDTO/page.mdx)
- [CartLineItemDTO](../../fulfillment/interfaces/fulfillment.CartLineItemDTO/page.mdx)
- [CartLineItemTotalsDTO](../../fulfillment/interfaces/fulfillment.CartLineItemTotalsDTO/page.mdx)
- [CartShippingMethodDTO](../../fulfillment/interfaces/fulfillment.CartShippingMethodDTO/page.mdx)
- [Context](../../fulfillment/interfaces/fulfillment.Context/page.mdx)
- [CreateCityGeoZoneDTO](../../fulfillment/interfaces/fulfillment.CreateCityGeoZoneDTO/page.mdx)
- [CreateCountryGeoZoneDTO](../../fulfillment/interfaces/fulfillment.CreateCountryGeoZoneDTO/page.mdx)
- [CreateFulfillmentAddressDTO](../../fulfillment/interfaces/fulfillment.CreateFulfillmentAddressDTO/page.mdx)
- [CreateFulfillmentDTO](../../fulfillment/interfaces/fulfillment.CreateFulfillmentDTO/page.mdx)
- [CreateFulfillmentItemDTO](../../fulfillment/interfaces/fulfillment.CreateFulfillmentItemDTO/page.mdx)
- [CreateFulfillmentLabelDTO](../../fulfillment/interfaces/fulfillment.CreateFulfillmentLabelDTO/page.mdx)
- [CreateFulfillmentSetDTO](../../fulfillment/interfaces/fulfillment.CreateFulfillmentSetDTO/page.mdx)
- [CreateGeoZoneBaseDTO](../../fulfillment/interfaces/fulfillment.CreateGeoZoneBaseDTO/page.mdx)
- [CreateProvinceGeoZoneDTO](../../fulfillment/interfaces/fulfillment.CreateProvinceGeoZoneDTO/page.mdx)
- [CreateServiceZoneDTO](../../fulfillment/interfaces/fulfillment.CreateServiceZoneDTO/page.mdx)
- [CreateShippingOptionDTO](../../fulfillment/interfaces/fulfillment.CreateShippingOptionDTO/page.mdx)
- [CreateShippingOptionRuleDTO](../../fulfillment/interfaces/fulfillment.CreateShippingOptionRuleDTO/page.mdx)
- [CreateShippingOptionTypeDTO](../../fulfillment/interfaces/fulfillment.CreateShippingOptionTypeDTO/page.mdx)
- [CreateShippingProfileDTO](../../fulfillment/interfaces/fulfillment.CreateShippingProfileDTO/page.mdx)
- [CreateZipGeoZoneDTO](../../fulfillment/interfaces/fulfillment.CreateZipGeoZoneDTO/page.mdx)
- [FilterableFulfillmentProps](../../fulfillment/interfaces/fulfillment.FilterableFulfillmentProps/page.mdx)
- [FilterableFulfillmentProviderProps](../../fulfillment/interfaces/fulfillment.FilterableFulfillmentProviderProps/page.mdx)
- [FilterableFulfillmentSetProps](../../fulfillment/interfaces/fulfillment.FilterableFulfillmentSetProps/page.mdx)
- [FilterableGeoZoneProps](../../fulfillment/interfaces/fulfillment.FilterableGeoZoneProps/page.mdx)
- [FilterableServiceZoneProps](../../fulfillment/interfaces/fulfillment.FilterableServiceZoneProps/page.mdx)
- [FilterableShippingOptionForContextProps](../../fulfillment/interfaces/fulfillment.FilterableShippingOptionForContextProps/page.mdx)
- [FilterableShippingOptionProps](../../fulfillment/interfaces/fulfillment.FilterableShippingOptionProps/page.mdx)
- [FilterableShippingOptionRuleProps](../../fulfillment/interfaces/fulfillment.FilterableShippingOptionRuleProps/page.mdx)
- [FilterableShippingOptionTypeProps](../../fulfillment/interfaces/fulfillment.FilterableShippingOptionTypeProps/page.mdx)
- [FilterableShippingProfileProps](../../fulfillment/interfaces/fulfillment.FilterableShippingProfileProps/page.mdx)
- [FindConfig](../../fulfillment/interfaces/fulfillment.FindConfig/page.mdx)
- [FulfillmentAddressDTO](../../fulfillment/interfaces/fulfillment.FulfillmentAddressDTO/page.mdx)
- [FulfillmentDTO](../../fulfillment/interfaces/fulfillment.FulfillmentDTO/page.mdx)
- [FulfillmentItemDTO](../../fulfillment/interfaces/fulfillment.FulfillmentItemDTO/page.mdx)
- [FulfillmentLabelDTO](../../fulfillment/interfaces/fulfillment.FulfillmentLabelDTO/page.mdx)
- [FulfillmentProviderDTO](../../fulfillment/interfaces/fulfillment.FulfillmentProviderDTO/page.mdx)
- [FulfillmentSetDTO](../../fulfillment/interfaces/fulfillment.FulfillmentSetDTO/page.mdx)
- [GeoZoneDTO](../../fulfillment/interfaces/fulfillment.GeoZoneDTO/page.mdx)
- [IBigNumber](../../fulfillment/interfaces/fulfillment.IBigNumber/page.mdx)
- [IFulfillmentModuleService](../../fulfillment/interfaces/fulfillment.IFulfillmentModuleService/page.mdx)
- [IMessageAggregator](../../fulfillment/interfaces/fulfillment.IMessageAggregator/page.mdx)
- [IModuleService](../../fulfillment/interfaces/fulfillment.IModuleService/page.mdx)
- [JoinerServiceConfig](../../fulfillment/interfaces/fulfillment.JoinerServiceConfig/page.mdx)
- [JoinerServiceConfigAlias](../../fulfillment/interfaces/fulfillment.JoinerServiceConfigAlias/page.mdx)
- [LineItemAdjustmentDTO](../../fulfillment/interfaces/fulfillment.LineItemAdjustmentDTO/page.mdx)
- [LineItemTaxLineDTO](../../fulfillment/interfaces/fulfillment.LineItemTaxLineDTO/page.mdx)
- [MessageAggregatorFormat](../../fulfillment/interfaces/fulfillment.MessageAggregatorFormat/page.mdx)
- [OrderAddressDTO](../../fulfillment/interfaces/fulfillment.OrderAddressDTO/page.mdx)
- [OrderAdjustmentLineDTO](../../fulfillment/interfaces/fulfillment.OrderAdjustmentLineDTO/page.mdx)
- [OrderChangeActionDTO](../../fulfillment/interfaces/fulfillment.OrderChangeActionDTO/page.mdx)
- [OrderChangeDTO](../../fulfillment/interfaces/fulfillment.OrderChangeDTO/page.mdx)
- [OrderClaimDTO](../../fulfillment/interfaces/fulfillment.OrderClaimDTO/page.mdx)
- [OrderCreditLineDTO](../../fulfillment/interfaces/fulfillment.OrderCreditLineDTO/page.mdx)
- [OrderDTO](../../fulfillment/interfaces/fulfillment.OrderDTO/page.mdx)
- [OrderExchangeDTO](../../fulfillment/interfaces/fulfillment.OrderExchangeDTO/page.mdx)
- [OrderItemDTO](../../fulfillment/interfaces/fulfillment.OrderItemDTO/page.mdx)
- [OrderLineItemAdjustmentDTO](../../fulfillment/interfaces/fulfillment.OrderLineItemAdjustmentDTO/page.mdx)
- [OrderLineItemDTO](../../fulfillment/interfaces/fulfillment.OrderLineItemDTO/page.mdx)
- [OrderLineItemTaxLineDTO](../../fulfillment/interfaces/fulfillment.OrderLineItemTaxLineDTO/page.mdx)
- [OrderLineItemTotalsDTO](../../fulfillment/interfaces/fulfillment.OrderLineItemTotalsDTO/page.mdx)
- [OrderReturnItemDTO](../../fulfillment/interfaces/fulfillment.OrderReturnItemDTO/page.mdx)
- [OrderShippingMethodAdjustmentDTO](../../fulfillment/interfaces/fulfillment.OrderShippingMethodAdjustmentDTO/page.mdx)
- [OrderShippingMethodDTO](../../fulfillment/interfaces/fulfillment.OrderShippingMethodDTO/page.mdx)
- [OrderShippingMethodTaxLineDTO](../../fulfillment/interfaces/fulfillment.OrderShippingMethodTaxLineDTO/page.mdx)
- [OrderSummaryDTO](../../fulfillment/interfaces/fulfillment.OrderSummaryDTO/page.mdx)
- [OrderTaxLineDTO](../../fulfillment/interfaces/fulfillment.OrderTaxLineDTO/page.mdx)
- [OrderTransactionDTO](../../fulfillment/interfaces/fulfillment.OrderTransactionDTO/page.mdx)
- [RestoreReturn](../../fulfillment/interfaces/fulfillment.RestoreReturn/page.mdx)
- [ReturnDTO](../../fulfillment/interfaces/fulfillment.ReturnDTO/page.mdx)
- [ServiceZoneDTO](../../fulfillment/interfaces/fulfillment.ServiceZoneDTO/page.mdx)
- [ShippingMethodAdjustmentDTO](../../fulfillment/interfaces/fulfillment.ShippingMethodAdjustmentDTO/page.mdx)
- [ShippingMethodTaxLineDTO](../../fulfillment/interfaces/fulfillment.ShippingMethodTaxLineDTO/page.mdx)
- [ShippingOptionDTO](../../fulfillment/interfaces/fulfillment.ShippingOptionDTO/page.mdx)
- [ShippingOptionRuleDTO](../../fulfillment/interfaces/fulfillment.ShippingOptionRuleDTO/page.mdx)
- [ShippingOptionTypeDTO](../../fulfillment/interfaces/fulfillment.ShippingOptionTypeDTO/page.mdx)
- [ShippingProfileDTO](../../fulfillment/interfaces/fulfillment.ShippingProfileDTO/page.mdx)
- [SoftDeleteReturn](../../fulfillment/interfaces/fulfillment.SoftDeleteReturn/page.mdx)
- [TaxLineDTO](../../fulfillment/interfaces/fulfillment.TaxLineDTO/page.mdx)
- [UpdateCityGeoZoneDTO](../../fulfillment/interfaces/fulfillment.UpdateCityGeoZoneDTO/page.mdx)
- [UpdateCountryGeoZoneDTO](../../fulfillment/interfaces/fulfillment.UpdateCountryGeoZoneDTO/page.mdx)
- [UpdateFulfillmentDTO](../../fulfillment/interfaces/fulfillment.UpdateFulfillmentDTO/page.mdx)
- [UpdateFulfillmentSetDTO](../../fulfillment/interfaces/fulfillment.UpdateFulfillmentSetDTO/page.mdx)
- [UpdateGeoZoneBaseDTO](../../fulfillment/interfaces/fulfillment.UpdateGeoZoneBaseDTO/page.mdx)
- [UpdateProvinceGeoZoneDTO](../../fulfillment/interfaces/fulfillment.UpdateProvinceGeoZoneDTO/page.mdx)
- [UpdateServiceZoneDTO](../../fulfillment/interfaces/fulfillment.UpdateServiceZoneDTO/page.mdx)
- [UpdateShippingOptionDTO](../../fulfillment/interfaces/fulfillment.UpdateShippingOptionDTO/page.mdx)
- [UpdateShippingOptionRuleDTO](../../fulfillment/interfaces/fulfillment.UpdateShippingOptionRuleDTO/page.mdx)
- [UpdateShippingOptionTypeDTO](../../fulfillment/interfaces/fulfillment.UpdateShippingOptionTypeDTO/page.mdx)
- [UpdateShippingProfileDTO](../../fulfillment/interfaces/fulfillment.UpdateShippingProfileDTO/page.mdx)
- [UpdateZipGeoZoneDTO](../../fulfillment/interfaces/fulfillment.UpdateZipGeoZoneDTO/page.mdx)
- [UpsertServiceZoneDTO](../../fulfillment/interfaces/fulfillment.UpsertServiceZoneDTO/page.mdx)
- [UpsertShippingOptionDTO](../../fulfillment/interfaces/fulfillment.UpsertShippingOptionDTO/page.mdx)
- [UpsertShippingOptionTypeDTO](../../fulfillment/interfaces/fulfillment.UpsertShippingOptionTypeDTO/page.mdx)
- [UpsertShippingProfileDTO](../../fulfillment/interfaces/fulfillment.UpsertShippingProfileDTO/page.mdx)

## Type Aliases

- [BigNumberRawValue](../../fulfillment/types/fulfillment.BigNumberRawValue/page.mdx)
- [BigNumberValue](../../fulfillment/types/fulfillment.BigNumberValue/page.mdx)
- [CalculatedShippingOptionPrice](../../fulfillment/types/fulfillment.CalculatedShippingOptionPrice/page.mdx)
- [CartPropsForFulfillment](../../fulfillment/types/fulfillment.CartPropsForFulfillment/page.mdx)
- [ChangeActionType](../../fulfillment/types/fulfillment.ChangeActionType/page.mdx)
- [Constructor](../../fulfillment/types/fulfillment.Constructor/page.mdx)
- [CreateGeoZoneDTO](../../fulfillment/types/fulfillment.CreateGeoZoneDTO/page.mdx)
- [Event](../../fulfillment/types/fulfillment.Event/page.mdx)
- [EventMetadata](../../fulfillment/types/fulfillment.EventMetadata/page.mdx)
- [ExpandScalar](../../fulfillment/types/fulfillment.ExpandScalar/page.mdx)
- [FilterQuery](../../fulfillment/types/fulfillment.FilterQuery/page.mdx)
- [FilterQueryProperties](../../fulfillment/types/fulfillment.FilterQueryProperties/page.mdx)
- [FilterValue](../../fulfillment/types/fulfillment.FilterValue/page.mdx)
- [FilterValue2](../../fulfillment/types/fulfillment.FilterValue2/page.mdx)
- [FindConfigOrder](../../fulfillment/types/fulfillment.FindConfigOrder/page.mdx)
- [GeoZoneType](../../fulfillment/types/fulfillment.GeoZoneType/page.mdx)
- [JoinerRelationship](../../fulfillment/types/fulfillment.JoinerRelationship/page.mdx)
- [LinkModulesExtraFields](../../fulfillment/types/fulfillment.LinkModulesExtraFields/page.mdx)
- [Message](../../fulfillment/types/fulfillment.Message/page.mdx)
- [MetadataType](../../fulfillment/types/fulfillment.MetadataType/page.mdx)
- [ModuleJoinerConfig](../../fulfillment/types/fulfillment.ModuleJoinerConfig/page.mdx)
- [ModuleJoinerRelationship](../../fulfillment/types/fulfillment.ModuleJoinerRelationship/page.mdx)
- [OperatorMap](../../fulfillment/types/fulfillment.OperatorMap/page.mdx)
- [OrderChangeStatus](../../fulfillment/types/fulfillment.OrderChangeStatus/page.mdx)
- [OrderClaimType](../../fulfillment/types/fulfillment.OrderClaimType/page.mdx)
- [OrderStatus](../../fulfillment/types/fulfillment.OrderStatus/page.mdx)
- [PrevLimit](../../fulfillment/types/fulfillment.PrevLimit/page.mdx)
- [Primary](../../fulfillment/types/fulfillment.Primary/page.mdx)
- [Query](../../fulfillment/types/fulfillment.Query/page.mdx)
- [RawMessageFormat](../../fulfillment/types/fulfillment.RawMessageFormat/page.mdx)
- [ReadonlyPrimary](../../fulfillment/types/fulfillment.ReadonlyPrimary/page.mdx)
- [ReturnStatus](../../fulfillment/types/fulfillment.ReturnStatus/page.mdx)
- [RuleOperatorType](../../fulfillment/types/fulfillment.RuleOperatorType/page.mdx)
- [Scalar](../../fulfillment/types/fulfillment.Scalar/page.mdx)
- [ShippingOptionPriceType](../../fulfillment/types/fulfillment.ShippingOptionPriceType/page.mdx)
- [StockLocationAddressDTO](../../fulfillment/types/fulfillment.StockLocationAddressDTO/page.mdx)
- [StockLocationDTO](../../fulfillment/types/fulfillment.StockLocationDTO/page.mdx)
- [UpdateGeoZoneDTO](../../fulfillment/types/fulfillment.UpdateGeoZoneDTO/page.mdx)
- [ValidateFulfillmentDataContext](../../fulfillment/types/fulfillment.ValidateFulfillmentDataContext/page.mdx)
